Author's response to reviews

Title: Buccal Soft Tissue Lipoma In An Adult Nigerian: A Case Report and Literature Review.

Authors:

Adeyi A Adoga (adeyiadoga@gmail.com)
Tonga L Nimkur (nimkuronga@yahoo.com)
Agabus N Manasseh (nanfwang@hotmail.com)
Godwin O Echejoh (ogechejoh@yahoo.com)

Version: 3 Date: 7 July 2008

Author's response to reviews:

Jos University Teaching Hospital
Otorhinolaryngology Unit
PMB 2076
Jos
Plateau state
Nigeria.
7th July 2008.

To: The Editor
Journal of Medical Case Reports

Dear Sir,

SUBMISSION OF REVISED MANUSCRIPT

I hereby submit a revised copy of our manuscript titled "BUCCAL SOFT TISSUE LIPOMA IN AN ADULT NIGERIAN: A CASE REPORT AND LITERATURE REVIEW".

The following corrections have been made to the original manuscript;

1. The pictures have been cropped
   a. the surgeons’ bloody gloves have been cut off showing just the tumor and mouth.
   b. gross picture showing just the lesion.
   c. microscopic picture showing the lesion which has also been lightened.

I hope our article is duly considered.

Thank you.

Dr. Adeyi A. Adoga.